Author's response to reviews

Title: Practice patterns of doctors of chiropractic with a pediatric diplomate: a cross-sectional survey

Authors:

Katherine A Pohlman (katherine.pohlman@palmer.edu)
Maria A Hondras (maria.hondras@palmer.edu)
Cynthia R Long (cyndy.long@palmer.edu)
Andrea G Haan (andrea.haan@palmer.edu)

Version: 3 Date: 19 May 2010

Author's response to reviews:

Dear Nina Titmus and the BioMed Central Editorial Team,

I am pleased to submit the final manuscript “Practice patterns of doctors of chiropractic with a pediatric diplomate: a cross-sectional survey” to BioMed Central’s Complementary and Alternative Medicine (BMC CAM).

In the editor’s comments listed below I addressed each concern on a point-by-point basis. I would like to sincerely thank the editor(s) for their thoughtful and insightful comments.

Sincerely,

Katherine A Pohlman, DC, DICCP

Formatting changes to be made:
----------------------------------

Major revisions (we require the author to make these changes)

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.

WE HAVE EMBEDDED THE TABLES INTO THE MAIN MANUSCRIPT.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Coloured text/underlining/track changes: please remove.

REMOVED.
Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figur

CROPPED TO MINIMIZE THE WHITE SPACE.

Tables: please remove the visible vertical lines from your tables

REMOVED THE VERTICAL LINES.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

ALL AUTHORS REVIEWED THE MANUSCRIPT AND MADE MINOR CHANGES TO INCREASE THE READABILITY.